Author's response to reviews

Title: Clinical and electrophysiological features of post-traumatic Guillain-Barré syndrome

Authors:

Xiaowen Li (lixiaow2005@sina.com)
Jinting Xiao (Xiaojinting1103@163.com)
Yanan Ding (dingyanan911@163.com)
Jing Xu (xujing301@sina.cn)
Chuanxia Li (lcx880806@163.com)
Yating He (361192703@qq.com)
Hui Zhai (zhaihui2368@sina.com)
Bingdi Xie (xbdhr@sina.com)
Junwei Hao (hjw@tmu.edu.cn)

Version: 3 Date: 12 May 2017

Author’s response to reviews:

12/5/2017

Tjalf Ziemssen
Editorial Office
BMC Neurology

Re: Manuscript # NURL-D-16-00516R2

Dear Editor:

Thank you for your letter sent on 21/4/2017 regarding our manuscript “Clinical and electrophysiological features of post-traumatic Guillain-Barré syndrome”.

Guided by the editor comments, we had found the reports and the names of the reviewers, Kalliopi Pitarokoili (Reviewer #1) and Shahriar Nafissi (Reviewer #2). The online peer review system made it very convenient. According to the Reviewer’s suggestion, we had invited a native
speaker to help us improve the manuscript again. Moreover, we checked the manuscript carefully again.

With these appropriate revisions, we believe that we have thoroughly addressed the reviewers’ concerns.

Below, please find a point-by-point response to reviewers’ comments.

We thank you for your time and consideration in re-evaluating the manuscript.

Sincerely,

Junwei Hao, PhD.
(On behalf of all authors)

Point-by-point reply to the Reviewers’ comments

Editor Comments:

BMC Neurology operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

Reply: We thank the editor for his advice. Guided by the editor comments, we had found the reports and the names of the reviewers, Kalliopi Pitarokoili (Reviewer #1) and Shahriar Nafissi (Reviewer #2). The online peer review system made it very convenient. We thank the two reviewers very much.

Kalliopi Pitarokoili (Reviewer #1):

The author have revised the mentioned points and performed the necessary language editing.

I have no further comments and I recommend the manuscript for publication.

Reply: We thank the reviewer very much.

Shahriar Nafissi (Reviewer #2):

Needs some language corrections before being published.
The manuscript has significantly improved after two revisions and I have no further comments.

Reply: We thank the reviewer very much. According to the reviewer’s suggestion, we have invited a native speaker to help us improve the manuscript again.